L2 Global Update

*only small changes (internal) since
last update.

*started looking at output and found
problems with CPS input so

attention went there.



L2 CPS Status

*symptom of trouble was phi

distribution

L2CPS.iphibin {abs{MCA IPET.in eta)=1.2} htcmp
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Mean = 10.8%
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L2 CPS Algorithm
Debugging Progress

*found a few problems

* calculation of position of center of
clusters from inputs was done
slightly wrong

*matching between layers had edge
effect problems

*phi wraparound was not taken
into account

* things look much better now
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